Author's response to reviews

Title: A survey of physicians knowledge regarding awareness of maternal alcohol use and the diagnosis of FAS.

Authors:

Alexandra C Nevin (Nevin@utoronto.ca)
Parshuram Christopher (c.parshuram@sickkids.ca)
Nulman Irena (inulman@sickkids.ca)
Koren Gideon (gkoren@sickkids.ca)
Einarson Adrienne (einarson@sickkids.ca)

Version: 2 Date: 18 Jan 2002

Reviewer #1
1. I have attempted to describe the results in a more clear and concise way
2. I included the statistical analysis in the beginning of the results
3. The TWEAK test is table 1 in the methods section and it is discussed in the discussion section with a reference
4. (Table 4) Respondents were asked to identify the 3 most important factors I added this to the table
5. I agree that table 5 is a little confusing, so I have deleted it and reported in the results section the % o physicians who identified diagnosis in the first year of life as determining the quality of life for the affected individual as being the single most important factor.
6. Actually the response rate is relatively high for these type of surveys and should not reflect a response bias, which is considered to be below 60%
7. The reference missing, is actually a typo

Reviewer#2
1. Initially we did try to survey pediatricians, but received too few responses to make a useful analysis of the results. The family physicians were much better responders so we went with their results alone. I was not sure if I should put this in the paper so I left it out?
2. We decided after much discussion to limit the paper to FAS only, a separate survey could be carried out at a later date to assess physicians knowledge of ARBD/ARND (alcohol related birth defects/alcohol related neurodevelopmental disorder) aka-FAE
3. I believe we made suggestions in the discussion section on how to improve physicians awareness in this field, but I have expanded it.